Author's response to reviews

Title: A case study using The United Republic of Tanzania: Costing nationwide HPV vaccine delivery using the WHO Cervical Cancer Prevention and Control Costing Tool

Authors:

Raymond Hutubessy (hutubessyr@who.int)
Ann Levin (annlevin@verizon.net)
Susan Wang (wangsu@who.int)
Winthrop Morgan (win@winthropsmorgan.com)
Mariam Ally (mariammwakobe@yahoo.com)
Theopista John (johnt@tz.afro.who.int)
Nathalie Broutet (broutetn@who.int)

Version: 5 Date: 9 October 2012

Author's response to reviews:

Dear Editorial Production team,

Herewith I submit the revised version including your formatting changes proposed of the manuscript entitled "A case study using the United Republic of Tanzania: Costing nationwide HPV vaccine delivery using the WHO Cervical Cancer Prevention and Control Costing Tool".

Sincerely

Raymond Hutubessy
WHO Initiative for Vaccine Research
Geneva, Switzerland